



No. 2027 August 2024

Evaluating the impact of export finance support on firm-level export performance: Evidence from Pakistan

Fabrice Defever Alejandro Riaño Gonzalo Varela



THE LONDON SCHOOL OF ECONOMICS AND POLITICAL SCIENCE



Economic and Social Research Council

Abstract

This paper evaluates the impact of two large export finance support schemes on firm-level export performance. The Export Finance Scheme (EFS) and the Long-Term Finance Facility for Plant & Machinery (LTFF), provide loans at subsidized interest rates for Pakistani exporters to finance working capital and the purchase of machinery and equipment respectively. We combine customs data with information on firms' participation in each program between 2015 and 2017 and use matching combined with difference-in-differences to estimate the effect of the subsidies on firms' export values, the number of products exported and the number of destinations they serve. We find that both programs deliver a large and positive impact on export growth rates—primarily along the intensive margin—and do so in an effective way relative to the direct financial cost of the subsidies.

Keywords: trade finance; export subsidies; working capital; machinery and equipment; export margins; Pakistan.

JEL: G21; F13; F14; F61; F65.

This paper was produced as part of the Centre's Trade Programme. The Centre for Economic Performance is financed by the Economic and Social Research Council.

We would like to thank Peter Morrow (the Editor), two anonymous referees, Farooq Arby, Ali Choudhary, Banu Demir, Andreas Eberhard, Sourafel Girma, Syed Samar Husnain, Dani Rodrik, Tim Schmidth-Eisenlohr and participants at the seminar at the University of Tübingen, the CESifo and LINER-AUEB conference on the role of institutions and policies in firm exporting, the World Congress of the International Economic Association in Medellín and the 2nd CITP Conference on Inclusive Trade Policy for helpful comments that have greatly improved the paper. All remaining errors are our own. Juan Pedro Gambetta provided excellent research assistance. The findings, interpretations, and conclusions expressed in this paper are entirely those of the authors. They do not represent the views of the World Bank, its Executive Directors, or the countries they represent.

Fabrice Defever, University of Lille, CNRS, IESEG and Centre for Economic Performance at London School of Economics. Alejandro Riaño, City University of London, GEP and CESifo. Gonzalo Varela, The World Bank.

Published by Centre for Economic Performance London School of Economic and Political Science Houghton Street London WC2A 2AE

All rights reserved. No part of this publication may be reproduced, stored in a retrieval system or transmitted in any form or by any means without the prior permission in writing of the publisher nor be issued to the public or circulated in any form other than that in which it is published.

Requests for permission to reproduce any article or part of the Working Paper should be sent to the editor at the above address.

© F. Defever, A. Riaño and G. Varela, submitted 2024.

1 Introduction

The global financial crisis of 2008 has forcefully demonstrated that access to finance is vital for firms to survive and grow in international markets.¹ In the short run, longer lags between production and payment make exporters more vulnerable to liquidity shocks and defaults, and thus render them more reliant on working capital financing than domestic firms (Amiti and Weinstein, 2011; Manova, 2013; Feenstra et al., 2014; Paravisini et al., 2015). At the same time, large adjustment costs and credit constraints can hinder exporters' ability to invest in physical capital, stunting their growth prospects in the long run (Riaño, 2011; Rho and Rodrigue, 2016; Brooks and Dovis, 2020; Leibovici, 2021; Kohn et al., 2022b). According to the World Economic Forum (2016), trade finance is one of the top three obstacles for exporters in developing countries.

Governments around the world have a long history of providing credit to exporters at subsidized interest rates to mitigate the financial frictions that affect international trade (Fleisig and Hill, 1984).² While developed countries now rely more on other instruments such as export credit guarantees (Moser et al., 2008; Felbermayr and Yalcin, 2013), direct subsidized credit to exporters remains popular among developing countries.³

In this paper we estimate the impact of subsidized loans—to finance both short-term working capital needs and long-term investment in machinery and equipment—on firm-level export performance. To do so, we analyze two large export finance support schemes offered by the State Bank of Pakistan (Pakistan's central bank, SBP hereafter)—The Export Finance Scheme (EFS) and the Long-Term Finance Facility for Plant & Machinery (LTFF) between

¹See e.g. Chor and Manova (2012), Bricongne et al. (2012), Paravisini et al. (2015).

²Fleisig and Hill (1984) report that in 1978, Canada, France, Germany, Italy, Japan, the United Kingdom, and the United States provided direct subsidized export credit worth 55 billion US dollars.

³For instance, the central bank of Bangladesh maintains an Export Development Fund of 3 billion US dollars that intends to facilitate access to financing in foreign exchange at subsidized interest rates for input procurement by manufacturing exporters (WTO, 2019). The Interest Equalisation Scheme on Pre and Post Shipment Rupee Export Credit offered by the government of India allows manufacturing exporters an interest subsidy of 3% on pre-and-post-shipment rupee credit for exports of 416 products. The central bank of Turkey's rediscount credit program is a pre and post shipment export financing facility that provides subsidized credit to exporters at low interest rates with little collateral requirement (Akgündüz et al., 2018), to name but a few examples.

2015 and 2017. EFS allows exporters to borrow funds over a period of up to 180 days to finance working capital at an interest rate 7 percentage points lower than the average lending rate during our period of analysis. LTFF, in turn, is targeted towards the purchase of machinery and equipment, offering exporters loans denominated in local currency at a fixed interest rate of 6% per annum over the duration of the credit (i.e. 3 percentage points below the average lending rate), with a maturity of up to 10 years.

A striking feature of these subsidy programs is their sheer scale. The average value of loans provided by EFS over our period of analysis was 3.66 billion US dollars per annum—or 16.75% of Pakistan's total exports—while loans financed by LTFF amounted to 263 million US dollars per annum, or 1.2% of the country's exports. These figures are orders of magnitude larger than the expenditure of most subsidies and other policy instruments analyzed in the literature that evaluates the impact of export promotion policies (see e.g. Volpe Martincus and Carballo, 2008; Görg et al., 2008; Volpe Martincus and Carballo, 2010a; Cadot et al., 2015; Van Biesebroeck et al., 2015, 2016; Munch and Schaur, 2018; Defever et al., 2019, 2020; Chávez et al., 2020). Thus, our work provides a unique perspective to examine how a large-scale trade policy shapes export performance at the firm level.

Pakistan offers a especially suitable environment to investigate the consequences of providing subsidized credit to exporters. After undertaking major trade liberalization reforms in the 1990s, its trade policy stance has significantly turned inwards in the last two decades. High levels of protectionism have produced lacklustre export performance relative to neighboring countries (Reis and Taglioni, 2013), while limited support for export promotion and political favoritism in the allocation of credit have lowered firms' productivity and increased barriers for firms to export (Khwaja and Mian, 2005; Zia, 2008; World Bank, 2021). Our work sheds light on the role that lowering the cost of short and long-term financing can play in alleviating distortions that hinder firms' export performance. These lessons are not only relevant for Pakistan, but also for other developing countries struggling to improve the dynamism of their exporters. Both schemes operate in two stages: first, a commercial bank screens a firm's loan applications and decides whether to grant or not the credit it requested; upon approval of this first stage, the commercial bank submits an application to refinance the loan with SBP at a lower interest rate. Approval of the application by SBP in the second stage determines both the interest rate that the firm pays to the commercial bank—which is lower than market prevailing rates—providing the credit and the refinancing interest rate at which the bank borrows from SBP. If an application is rejected by SBP in the second stage, the firm can still borrow from the commercial bank, but at a higher interest rate.

We estimate the effect of using EFS and LTFF on firm-level export performance using a range of matching estimators combined with difference-in-differences to account for the non-random selection of firms in taking advantage of the schemes. Our empirical strategy is underpinned by two identification assumptions. The first is that once we control for a broad range of characteristics observed prior to a firm using a scheme, we can ascribe any differences in performance between treated and control firms to the effect of the subsidies. The variables we include in the estimation of the treatment selection model are firms' export sales, the number of products exported and destination market served, number of foreign buyers that firms sell to (all in levels and growth rates), importing status, and main export productdestination fixed effects. Our objective is to control for any factors that simultaneously affect the likelihood of firms using the subsidies and their export performance. Putting it differently, we try to ensure that the estimation of the effects of the subsidy is based on the comparison of firms undergoing similar pre-treatment export growth trends.

The second identification assumption is that firms with a similar export trajectory can, nevertheless, experience different treatment status due to factors that are not correlated with their export performance. Anecdotal evidence suggests that this can be the case because creditworthy exporters are, not infrequently, rejected by SBP in the second stage of their application. Interviews with SBP officials indicate that the most common reason for these rejections is that commercial banks have reached a specific refinancing limit set by SBP in an opaque way, and which—notably—appears to be unknown to the bank's managers.

We find that both subsidy programs—but particularly LTFF, which incentivizes the purchase of machinery and equipment—have a large and positive effect on the export performance of recipient firms. More precisely, we estimate that the growth rate of export sales for firms taking advantage of LTFF is 20-31 percentage points higher than if they had not used the subsidy, while EFS, the program subsidizing working capital, increases the growth of export sales by 11.5-14.3 percentage points relative to counterfactual. Our results also show that both subsidies have a significant, albeit smaller, positive effect on growth along the extensive margin (i.e. in terms of the number of products a firm exports and the number of destinations it serves). The large effects we estimate are consistent with a broad literature in international trade and macroeconomics showing that a lower cost of working capital and cheaper financing of machinery and equipment allow firms to increase the scale of their operation (Manova, 2013; Feenstra et al., 2014) and achieve this transition more rapidly (Rho and Rodrigue, 2016; Kohn et al., 2022a), as well as with evidence suggestive of a high elasticity of exports with respect to changes in trade finance (Zia, 2008; Paravisini et al., 2015; Demir et al., 2017; Akgündüz et al., 2018).

This paper contributes to several strands of research. In evaluating the effects of two policy instruments that subsidize distinct margins of firms' financing needs, our paper provides an overarching perspective about the critical role that the cost of finance plays in shaping export performance and speaks to the extensive literature studying the role of financial factors and credit constraints in international trade (e.g. Greenaway et al., 2007; Amiti and Weinstein, 2011; Manova, 2013; Feenstra et al., 2014; Paravisini et al., 2015; Muûls, 2015). Our analysis of a subsidy to the purchase of machinery and equipment complements the body of work in macroeconomics examining how credit frictions affect firms' capital accumulation and export dynamics (Caggese and Cuñat, 2013; Kohn et al., 2016; Brooks and Dovis, 2020; Kohn et al., 2022b). Lastly, our work is closely related to Zia (2008) and Akgündüz et al. (2018), who also investigate the consequences of subsidizing working capital for exporters in Pakistan and Turkey respectively. We add to the latter by investigating the impact of incentives to long-term investment and analyzing the consequences of these subsidies for firms' export diversification along the extensive margin.

The rest of the paper is structured as follows: Section 2 describes the two export finance support schemes we evaluate. Section 3 introduces the data we employ and provides summary statistics on firm-level export performance and usage of export finance support schemes. Section 4 describes our empirical strategy. Section 5 presents our estimates and Section 6 presents a back-of-the-envelope cost-benefit analysis of both schemes; Section 7 concludes.

2 Export Finance Support Schemes

This section describes the main features of the two export finance support schemes we evaluate in this paper: The Export Finance Scheme (EFS) and the Long-Term Finance Facility for Plant & Machinery (LTFF).

2.1 The Export Finance Scheme (EFS)

EFS was established in 1973 with the objective of increasing Pakistan's manufacturing exports. The scheme is currently available to firms exporting most manufacturing products, with the exception of 20 products included in a negative list (see online appendix A), which was last updated in 2011.

EFS offers short-term loans (with a maturity up to 180 days) to finance working capital for exporters at subsidized interest rates. More specifically, the scheme allows commercial banks to refinance loans they provide to exporters by borrowing from the State Bank of Pakistan (SBP) at below-market interest rates. Thus EFS provides incentives to commercial banks to finance the short-term capital needs of exporters by making liquid funds available to them at subsidized rates. It is important to note that SBP does *not* provide loans directly to exporters; commercial banks bear the credit risk if an exporter defaults on its loan. EFS specifies both the interest rate that the commercial bank charges to the exporter and the refinancing rate at which the commercial bank borrows from SBP, and therefore fixes the intermediation margin that the commercial bank earns by providing the loan to the exporter. Conversations with bank managers indicated that this margin is sufficiently attractive to motivate commercial banks to participate in the scheme.

The scheme operates in two stages summarized in Figure 1. First, a firm with an export order or letter of credit at hand approaches a commercial bank to request a working capital loan, and the bank evaluates the firm's request as it does any standard loan application. If the bank decides to extend credit to the exporter, then—in the second stage—the bank has the option to submit an application to SBP, supported by documentation provided by the exporter, to take advantage of the refinancing facility offered by EFS. If SBP approves the application submitted in the second stage, then the commercial bank obtains funds from SBP equal to the value of the loan it disbursed to the exporter.

Figure 1: Export finance support schemes—application timeline



Between 2015 and 2017, the interest rate charged to exporters borrowing under EFS was 2% per annum and the refinancing rate for commercial banks was 1%. Since the average market lending rate over the same period was 9%, the interest rate subsidy provided to exporters—i.e. the difference between the market lending rate and interest rate charged on

EFS loans—was 7 percentage points, a similar figure to the subsidization rate of 6 percentage points calculated by Zia (2008) during the 1990s and early 2000s.

2.2 The Long-Term Finance Facility for Plant & Machinery (LTFF)

The LTFF is a financing facility set up by SBP in 2007 with the objective of promoting export-led industrial growth. It offers subsidized loans in local currency with a maximum maturity of 10 years to export-oriented firms (i.e. firms that either export at least 50% of their sales, or have an export turnover of at least 5 million US dollars) to finance long-term investments in physical capital such as plant and machinery of up to 1.5 billion Pakistani rupees (approximately 9 million US dollars). In contrast to EFS, during our period of analysis LTFF was only available to firms operating in 20 sectors (see Table A.2 in online appendix A; the scheme became available to exporters in all sectors in January 2020).

LTFF also operates in two stages. First, a firm approaches a commercial bank to obtain a loan for the purchase of new machinery or equipment. Conditional on approving the credit, the bank then submits an application to SBP to refinance the loan by taking advantage of LTFF. Upon approval of the second stage by SBP, LTFF specifies the interest rate that the exporter pays for the loan and the refinancing rate for the commercial bank. During our period of analysis, the interest rate faced by firms was 6% per annum throughout the term of the loan and regardless of its maturity; the refinancing rate for commercial banks, on the other hand, decreased with the loan's maturity, from 4.5% per annum for loans up to 3 years to 3% for loans up to 10 years.

2.3 How are the Schemes Administered by the SBP?

The total allocation of funds for export finance support schemes is decided by SBP every fiscal year. Disbursement is carried out on a first-come, first-served basis subject to commercial banks not surpassing their specific refinancing limit, which is also set by SBP. Conversations with SBP officials revealed that bank limits are chosen on the basis of a wide range of factors including banks' rating given by SBP inspectors, volume of foreign exchange deposits, market share in trade and long-term financing, in addition to other unspecified considerations. Notably, when we asked commercial bank managers about their refinancing limits, not only they did not know what the specific limit for their own institution was, but each listed different criteria when we asked them about how SBP determined such limits!

Data on stage two rejections by SBP (unfortunately, only available to us for the period 2018-2020) show that a high number of firms—103 over these two years—are unable to use the subsidies after commercial banks have approved their credit and that a substantial share of these have taken advantage of the subsidies during our period of analysis.⁴ Interviews with SBP officials indicated that the main reason for rejecting applications in the second stage was that the commercial bank from which an exporter had requested credit had already reached its refinancing limit.⁵ The institutional design of the export finance schemes, therefore, provides ample scope for creditworthy exporters to not being able to take advantage of them.

3 Data and Summary Statistics

This section describes the data used in the empirical analysis. It provides summary statistics regarding firm-level export performance as well as participation in EFS and LTFF, and the value of loans outstanding financed by each scheme during our period of study.

We use two data sets in this paper. Customs data collected by the Federal Board of Revenue contain the universe of export and import transactions for firms in Pakistan over the period 2014-2017. These data have information on the value of firms' exports and imports by product at the HS 8-digit level as well as the country of origin and destination

⁴More specifically, three out of four firms rejected from the schemes between 2018-2020 had successfully obtained EFS funds during our period of analysis, while half of them had made use of LTFF.

⁵In this event, the exporter still has the option to obtain credit from the commercial bank, but paying a higher interest rate. Because of Pakistan's low level of financial development, it is unlikely that exporters would be able to request a loan from another bank that has not reached their refinancing limit (Khwaja and Mian, 2008; Zia, 2008).

of trade flows. For the years 2014 and 2015 we also observe the number of foreign buyers that exporters sell to. Between 2015 and 2017 there are 20,052 firms reporting at least one positive export transaction in at least one of 2,844 HS 8-digit products sold to 202 countries. The data on export finance support schemes provided by SBP includes information on which firms used EFS and LTFF and the value of the loans they received between 2015 and 2017.⁶ Both data sets are linked using firms' National Tax Number.

Table 1 provides a first look at the number of exporters and their performance over the period of analysis. The number of active exporters remains stable, with approximately 14,500 firms exporting each year. Table 1 reassuringly shows that firm-level export patterns in Pakistan are in line with figures for countries at a similar stage of development, as documented by Fernandes et al. (2016).

Table 1: Export patterns in Pakistan, 2015-2017

Year	#	Median exports	Mean exports	Mean # HS-8	Mean $\#$ destinations
	Exporters	per exporter	per exporter	per exporter	per exporter
2015	14,765	89.33	1,587	5.12	3.48
2016	$14,\!433$	87.45	1,468	5.12	3.50
2017	$14,\!536$	86.17	$1,\!445$	5.17	3.35

Export values are denominated in thousand US dollars.

We now turn to document the extent to which firms utilize export finance support schemes. The first two columns of Table 2 reveal that approximately 5% of exporters participate in EFS and fewer than 1% utilize LTFF in a given fiscal year. While the number of exporters using EFS remains roughly constant between 2015 and 2017, the number of exporters taking advantage from LTFF doubled, albeit from a much lower base.

The last four columns of Table 2 demonstrate the remarkable large scale of the financing made available by EFS and LTFF. Loans granted under EFS amount to 3.66 billion US

⁶It is important to note that we only observe the total value of loans obtained by a firm through each scheme in a given year. Thus, the data does not allow us to distinguish if loans are provided for certain export transactions and not others.

dollars per year on average, or 16.75% of Pakistan's exports between 2015 and 2017. Even accounting for its narrower scope and most recent implementation, LTFF finances loans worth 263 million US dollars per year on average, or 1.2% of total exports over the same period. To put these figures in context, the entire annual budget of export promotion agencies, including those of developed countries like Australia, Japan and the UK, does not exceed 500 million US dollars (Volpe Martineus, 2010).

Year	# o:	f exporters	Tota	al Value of	Value	e of loans /
	re	eceiving	Loans	outstanding	total	exports $(\%)$
	EFS	LTFF	EFS	LTFF	EFS	LTFF
2015	832	64	3.42	0.14	14.6	0.6
2016	812	80	3.77	0.22	17.8	1.0
2017	814	125	3.79	0.43	18.1	2.1

Table 2: Usage of export finance support schemes, 2015-2017

Total value of loans outstanding is denominated in billion US dollars.

4 Empirical Strategy

Our objective is to estimate the effect of EFS and LTFF on firm-level export performance, i.e. on firms' export sales, number of products exported, and number of destination countries served over the period 2015-2017. The main challenge we face is to credibly estimate what would have been the export outcomes of firms using export finance support schemes, had they not actually received these subsidies. It is unlikely that the performance of unsubsidized producers would provide an unbiased estimate of the average counterfactual outcomes for treated firms because the latter have deliberately chosen to participate in these programs. Following the literature that evaluates the effects of export promotion policies on firmlevel export performance (see e.g. Volpe Martincus and Carballo, 2008; Görg et al., 2008; Cadot et al., 2015; Van Biesebroeck et al., 2015; Munch and Schaur, 2018, among others), we use matching estimators combined with difference-in-differences to estimate the average treatment effect of participating in export finance support schemes.^{7,8}

We consider an exporting firm as being treated if it did *not* use a given export finance support scheme in 2015 (the first year in which we observe firms usage of the schemes), but did so in 2016 and/or 2017.⁹ There are two identification assumptions that underpin our estimation strategy of the average treatment effect of each scheme. The first, is that selection into treatment is fully determined by observable characteristics. This means that once we control for all relevant factors that determine firms' decision to use an export finance support scheme—observed in 2015, before the treatment takes place—we attribute any differences in export performance between treated firms and those in the comparison group to the respective subsidy.

The covariates we use to explain selection into treatment determine a comparison group of firms that shares the same observable characteristics as treated firms and is subject to similar shocks, but did not use export finance support. The control variables we include in our estimation include the share of exports accounted for by products eligible in each scheme, total export sales, the number of foreign buyers an exporter sells to, the number of products exported and destination markets reached (all in levels and growth rates between 2014 and 2015 to ensure that treated and controls firms have similar pre-treatment trends in export performance)¹⁰, an importing status indicator, and main export product-

⁷An alternative empirical strategy to estimate the effect of LTFF specifically, would have been to use a regression discontinuity design or intention-to-treat regression that exploited the 5 million US dollar turnover threshold required to participate in the scheme. Unfortunately, only one of the firms using LTFF has export sales below this threshold. Excluding all firms that do not use LTFF and have a turnover below 5 million US dollars from the estimation has a negligible impact on our matching-based estimates.

⁸Another alternative empirical strategy would have been to employ a difference-in-differences design using the exports of non-eligible products as a control group for the exports of eligible products. The main drawback of this strategy is that eligible products account for approximately 90% of exports for most firms that use EFS. Due to the small share of these products in treated firms' exports, non-eligible products display large swings in the annual growth rates of export outcomes. This makes it hard to justify the assumption that the performance of non-eligible products provides an adequate estimate for the counterfactual growth rate of exports of eligible products for treated firms.

⁹This entails excluding a substantial number of firms that use an export finance support scheme in every year of our period of analysis—a restriction that is most important for EFS.

¹⁰In online appendix C, we probe the assumption that conditional on covariates, treated and control firms were already experiencing different growth trajectories before the former receive subsidies.

destination fixed effects.¹¹ While we acknowledge that there is no 'perfect' solution to deal with confounding demand shocks, the inclusion of the aforementioned fixed effects restricts the comparison between subsidized and control firms to those sharing the same main export product-destination, and would therefore account for potential demand shocks that would affect both groups of firms in a similar way.

Larger firms are more likely to continue exporting, have a relationship with a commercial bank and use external finance to fund their working capital needs and investment in machinery and equipment (Beck et al., 2008; Blum et al., 2013). Similarly, firms that export a higher share of products eligible for a certain scheme are more likely to use the it. The inclusion of a broad range of pre-treatment export outcomes and main export-product-destination fixed effects implies that control firms not only have similar export performance across a broad range of dimensions as the subsidized firms they are compared against, but, crucially, also focus their exports in the same products and destination markets. This intends to allay concerns that our results could be contaminated by positive demand shocks that would simultaneously increase the likelihood of firms using export finance schemes and their export sales (Munch and Schaur, 2018). Nevertheless, it is worth noting that the presence of timevarying unobservables that affect both firms' usage of the schemes and export performance can bias the average treatment effects we estimate.

The second key identification assumption is that there are plausible factors, uncorrelated with export performance, that explain why firms with similar observable characteristics prior to treatment eventually end up with different treatment status (Imbens and Rubin, 2015; McKenzie, 2021). As we noted in Section 2.3, institutional features of the process through which SBP administers the export finance schemes offer a likely explanation for this event. The most common reason for SBP to reject applications of firms to use the subsidies is that commercial banks they are borrowing from have exceeded their refinancing limit; however, neither exporters nor their banks appear to know the likelihood of this occurrence. In the

¹¹More precisely, these are the interaction of two dummy variables indicating the most important HS 4-digit export product and export destination (in terms of value) for each firm.

end, it is likely that a non-negligible number of creditworthy exporters are not able to take advantage of the subsidies due to reasons that have nothing to do with their potential export performance.

We use the doubly-robust estimator proposed by Wooldridge (2007), which integrates inverse probability weighing with covariate adjustment, to implement different matching procedures. This estimator provides two opportunities to adjust for selection on observables, thus delivering unbiased inference of causal effects as long as either the conditional mean (outcome) regression or the selection-into-treatment models are correctly specified. The estimation proceeds in two steps; first, we estimate the probability that a firm is treated as a function of a vector of observable characteristics by means of a probit model— $\hat{P}(T_i = 1 | \mathbf{X}_i)$, where T_i is an indicator taking the value 1 when firm *i* is treated, \mathbf{X}_i is the vector of covariates measured in 2015 (including the main export product-destination fixed effects) and \hat{P} denotes the estimated propensity score. In the second stage we estimate outcome regressions of the following type:

$$g_i = \alpha + \beta T_i + \mathbf{X}_i' \gamma + \varepsilon_i, \tag{1}$$

for each measure of export performance and for each export finance support scheme.¹² The dependent variable g_i is the average growth rate of a given export performance outcome for firm *i* between 2015 and 2017, calculated using the mid-point annual growth rate between two years, *t* and t-1—i.e. $g_{i,t} = (y_{i,t} - y_{i,t-1}) / [0.5 \times (y_{i,t} + y_{i,t-1})]$.¹³ Estimating the outcome regression (1) in growth rates ensures that time-invariant factors that affect both the use of export finance support schemes and export performance are duly controlled for.

We use the propensity score estimated in the first stage to construct three different set of

¹²An important caveat of our analysis is that we evaluate the impact of each export finance support scheme independently, and therefore we are not able to determine whether there are synergies between the two programs in terms of their impact on export performance. Unfortunately, there are only a handful of firms in our data that use LTFF without using EFS; this precludes us from estimating the joint effect of both schemes, as Volpe Martineus and Carballo (2010b) do, for instance.

¹³Using the mid-point growth rate bounds the growth rate between -2 and +2, and thus limits the influence of large swings in export values.

weights for treated and untreated firms when we estimate (1); namely, (i) inverse probability (IPW), (ii) propensity score matching (PSM) and (iii) Mahalanobis or nearest neighbor matching (NNM). When using IPW, we give the weight $1/\hat{P}$ to treated firms and $1/(1-\hat{P})$ to control firms. Doing so gives greater importance to both treated firms that have a relatively low estimated probability of using a given subsidy and untreated firms that are more likely to use the same scheme based on their observed characteristics. Propensity score matching assigns a weight of 1 to every treated firm and its respective control, i.e. the untreated firm that is closest in terms of its propensity score, and 0 otherwise. Nearest-neighbor matching operates in the same way as PSM but treated and control firms are matched according to the Mahalanobis norm between covariates rather than according to their propensity score (once again, using only the closest neighbor for the match). Note that we always include the vector of covariates $\mathbf{X}_{\mathbf{i}}$ we used to estimate the propensity score in the outcome regression (1) too. We also ensure that the estimation of (1) only includes observations for which there is overlap in the distribution of the propensity score between treated and non-treated firms in order to satisfy the common support assumption.

5 Results

In this section we present the estimates of the effect of export finance support schemes on export performance. We begin by discussing the estimates of the model predicting the probability that an exporter uses a given subsidy program and assess the quality of the matching procedure. We then move to the estimates of the average treatment effect of EFS and LTFF on firm-level export outcomes.

Table 3 presents the estimates of the probit models used to calculate the propensity score for each subsidy. While the key objective of propensity score estimation is to achieve a balancing score—i.e. weighting the observations to eliminate biases in estimated treatment effects due to differences in the distribution of the baseline covariates—these estimates also

shed light on the forces that determine firms' participation in each export finance scheme.

The propensity score specification for each subsidy program predicts treatment status reasonably well, but crucially, a substantial share of the variation in firms' usage of export finance schemes remains unexplained. This is important because as Blundell and Costa Dias (2009) note, on one hand, if the treatment model predicts 'too well' then the distributions of propensity scores for treated and control firms do not overlap, making it hard to find suitable non-treated firms to use as comparison; on the other hand, the conditional independence assumption necessary to recover consistent average treatment effects is difficult to justify.

The results reported in Table 3 show that firm size is the main factor predicting firms' use of subsidies. Larger exporters not only have higher working capital and machinery and equipment needs but are also more likely to use commercial banks to finance these investments—a pre-requisite to take advantage of EFS and LTFF—rather than relying on other financing methods like trade credit provided by suppliers or retained earnings (Petersen and Rajan, 1997; Auboin, 2007). Other indicators of past export performance, for the most part, do not appear to have a strong impact on exporters' usage of subsidies. This suggests that firms that are growing faster between 2014 and 2015 are not, for the most part, more likely to use export performance growth trends prior to treatment remain non-significant in explaining firms' subsidy use after reweighting. We interpret these results as an indication that our choice of pre-treatment controls is effective in selecting comparison firms that are on a similar export performance trajectory as treated firms before they receive subsidies.

The inclusion of main export product-destination fixed effects in our estimating equations has two main implications. First, they improve the accuracy of prediction of the treatment model, particularly for EFS (estimates of treatment effects without fixed effects are presented in Tables B.4 and B.5 in the online appendix). Second, a substantial number of firms for which these fixed effects perfectly predict treatment status (which we denote as 'singletons') are dropped from the estimation. This is important, because by construction we are restricting the potential set of comparison firms to those for which their main export product-destination is such that there is a chance for firms to receive a given subsidy and therefore helps us to control for potential demand shocks in a given product-destination that would affect treated and control firms in a similar way.

	EFS	LTFF
	(1)	(2)
Share of exports in EFS	-0.096	
Negative list	(0.186)	
Share of exports in LTFF		2.040
Eligible list		(1.299)
Log export value	0.233^{***}	0.516^{***}
	(0.045)	(0.128)
Log # buyers	-0.034	-0.112
	(0.095)	(0.191)
Log # destinations	0.022	0.478^{**}
	(0.101)	(0.222)
Log # products	0.165^{**}	-0.096
	(0.080)	(0.159)
Δ Log export value	0.030	0.334^{*}
	(0.084)	(0.187)
$\Delta \text{ Log } \# \text{ buyers}$	-0.100	-0.429
	(0.127)	(0.272)
$\Delta \text{ Log } \# \text{ destinations}$	0.040	-0.149
	(0.138)	(0.321)
$\Delta \text{ Log } \# \text{ products}$	-0.059	0.690***
	(0.111)	(0.234)
Importer status	-0.074	0.230
	(0.126)	(0.328)
$HS4 \times destination fixed effects$	Y	es
Observations	8,135	$8,\!839$
Excluding singletons	2,027	745
Pseudo R-squared	0.225	0.515
Joint significance test (p-value)	0.00	0.00

Table 3: First-stage probit for the probability of participating in an export finance scheme

The table reports the coefficients of probit models estimated among the set of firms observed in 2015. The dependent variable in column (1) [(2)] takes the value 1 if a firm that did not participate in EFS [LTFF] in 2015 uses EFS [LTFF] in 2016 and/or 2017 and 0 otherwise. All covariates are measured in 2015 (Δ denotes the difference between the value of a given variable in 2015 relative to its value in 2014). Standard errors are in parenthesis. *, **, *** indicate significance at the 10%, 5% and 1% levels, respectively.

We next examine how the different matching procedures we employ perform in terms of achieving balance in the covariates used to predict treatment status. Table 4 presents standardized differences and variance ratios for all combinations of subsidy program and method of matching. Large differences in covariates in the raw data reinforce the notion that export outcomes of firms that did not use export finance support schemes do not provide an accurate estimate of the counterfactual outcome for treated firms. Table 4 shows that weighting substantially reduces the differences in the first and second moments of covariates determining the probability of treatment. In most cases, the standardized differences of covariates fall well below the 20% threshold criterion commonly employed in the literature on treatment effects (Görg et al., 2008; Caliendo and Kopeinig, 2008), and variance ratios tend to move closer to unity after weighting.

Table 5 presents the pseudo R-squared and joint significance tests obtained after reestimating the propensity score probit model using only treated firms and their respective controls (Caliendo and Kopeinig, 2008). The pseudo R-squared of these regressions falls substantially compared to the value of the same statistic reported in Table 3, indicating that weighting produces a notable improvement in the balance of pre-treatment covariates between the treated and comparison groups. While we clearly cannot reject the null hypothesis of joint insignificance of all covariates for EFS, the opposite result obtains for LTFF when using inverse probability and propensity score weighting.¹⁴

¹⁴Table B.3 in the online appendix shows that all covariates are jointly insignificant across all matching estimates when we do not include fixed effects. This result suggests that LTFF usage is strongly driven by firms exporting certain products to specific markets.

quality
matching
of
Indicators
4:
Table

RawPanel A: EFSShare of exports in EFS in Negative list-0.46Log export value 0.92 Log # buyersLog # destinationsLog # products 0.40	w IPW	PSM	NTN ATA	Ļ	TDIXT		NTN IN I
Panel A: EFSShare of exports in EFS in Negative listLog export valueLog # buyersLog # destinationsLog # productsLog # products		1	TNTINI NT	Kaw	LF W	NICH	TATTAT NT
Share of exports in EFS in Negative list -0.46 Log export value 0.92 Log # buyers 0.69 Log # destinations 0.63 Log # products 0.40							
Log export value 0.92 Log # buyers 0.69 Log # destinations 0.63 Log # products 0.40	16 -0.04	-0.18	-0.11	0.59	0.96	0.77	0.87
Log # buyers 0.69 Log # destinations 0.63 Log # products 0.40	2 -0.00	-0.09	0.34	0.59	0.62	0.70	0.74
Log # destinations 0.63 Log # products 0.40	· -0.00	0.12	0.26	0.88	0.67	0.73	1.00
$Log \ \# \ products $ 0.40	3 -0.02	0.15	0.14	1.05	0.80	0.97	1.02
	0.03	0.20	0.21	1.13	0.97	1.00	1.10
Δ Log export value -0.03	0.00	0.19	0.12	0.50	0.70	0.75	0.99
$\Delta \text{ Log } \# \text{ buyers}$ -0.07	00.0- 70	-0.03	0.07	0.77	1.00	0.72	1.37
$\Delta \log \#$ destinations 0.01	1 -0.02	-0.10	-0.02	1.01	1.19	1.02	1.55
$\Delta \log \#$ products -0.04	04 -0.05	-0.06	0.09	0.67	0.91	1.16	1.30
Importer status 0.37	37 -0.02	-0.07	0.16	1.18	1.00	1.00	1.04
Panel B: LTFF							
Share of exports of LTFF in eligible list -0.06)6 -0.10	-0.10	-0.01	1.17	1.21	1.29	0.99
Log export value 1.86	6 0.11	-0.10	0.72	0.43	1.06	0.72	0.73
Log # buyers 1.57	7 0.05	-0.23	0.51	0.89	1.29	0.94	1.26
Log # destinations 1.46	6 0.18	-0.21	0.56	0.87	1.18	1.05	1.29
$Log \ \# \ products $ 0.74	4 0.31	-0.15	0.38	0.99	1.10	1.39	1.12
Δ Log export value 0.01	0.12	0.12	0.31	0.32	0.55	0.26	0.81
$\Delta \operatorname{Log} \# \operatorname{buyers}$ -0.04)4 0.15	0.08	0.11	0.42	0.81	0.92	1.05
$\Delta \log \#$ destinations 0.09	0.07	0.14	0.20	0.45	1.18	1.33	1.09
$\Delta \log \# $ products 0.21	10.09	0.27	0.31	0.39	0.90	0.49	0.84
Importer status 1.25	5 0.01	0.06	0.28	0.22	0.95	0.81	0.44

of covariate X_k in the treatment and comparison groups respectively and $s_{k,1}^2$ and $s_{k,0}^2$ are the sample variances of covariate X_k in the treatment and comparison groups respectively. The variance ratio is defined as $VR = s_{k,1}^2/s_{k,0}^2$. IPW stands for inverse probability weighting, PSM stands for propensity score matching weighting and NNM for Mahalanobis matching weighting. The standardized difference for each covariate X_k is given by $SD_k = (\overline{X}_{k,1} - \overline{X}_{k,0})/\sqrt{(s_{k,1}^2 + s_{k,0}^2)}/2$, where $\overline{X}_{k,1}$ and $\overline{X}_{k,0}$ denote the sample mean

EFS		IPW	PSM	NNM
pseudo R-squared	0.22	0.01	0.12	0.07
Joint significance test (p-value)	0.00	1.00	1.00	1.00
LTFF		IPW	PSM	NNM
pseudo R-squared	0.52	0.08	0.20	0.23
Joint significance test (p-value)	0.00	0.02	0.00	0.98

Table 5: Joint significance and pseudo R-squared of treatment status model

The table reports the pseudo R-squared and the p-value associated with the chi-squared joint significance test from running the probit model of the probability of participating in each export finance scheme, and the same statistics when the model is estimated using only the treated and relevant comparison firms. IPW stands for inverse probability weighting, PSM stands for propensity score matching weighting and NNM for Mahalanobis matching weighting.

We now move to discuss the effect of EFS and LTFF on export outcomes. The sample we use for the estimation consists of firms that export in every year between 2014 and in 2017, and for which the main export product-destination fixed effects do not perfectly predict the usage of subsidies. Among the 2,027 firms we use to estimate the effect of EFS, 122 are treated (they do not use EFS in 2015 and either use it in 2106 and/or 2017); among the 745 firms included in the sample used to evaluate LTFF, 71 are treated. Since the distribution of propensity scores for treated and comparison firms exhibits full overlap for each subsidy, we do not exclude any treated firms of our estimation on this account.

Table 6 presents our estimates of the average treatment effect on the treated for EFS in terms of its impact on the growth rate firms' total value of exports, number of products exported (at the HS 8-digit level) and the number of countries to which a firm exports. OLS estimates reveal a positive and highly significant impact of EFS on export performance along both intensive and extensive margins, with the former experiencing a stronger response. The results reported in rows 2-4 of column (1) in Table 6 indicate that using EFS increases the annual growth rate of exports for participant firms between 11.5 and 14.3 percentage points. Controlling for the large degree on imbalance in firms' observable characteristics prior to using EFS (i.e. comparing the matching estimates with the unweighted OLS ones), reduces

	Export	#	#
	value	destinations	products
	(1)	(2)	(3)
OLS	0.174***	0.070***	0.027
	(0.038)	(0.025)	(0.025)
Inverse probability (IPW)	0.143^{***}	0.060^{**}	0.025
	(0.041)	(0.025)	(0.027)
Propensity score (PSM)	0.115^{***}	0.076^{***}	0.006
	(0.041)	(0.020)	(0.020)
Mahalanobis matching (NNM)	0.139^{***}	0.087^{***}	0.011
	(0.048)	(0.034)	(0.035)
Average growth rate			
of treated firms	0.042	0.044	-0.025

Table 6: Average treatment effect of Export Finance Scheme (EFS) on the average growth rate of export outcomes

Each entry in the table reports the average treatment effect on the treated firms that made use of EFS—i.e. the estimated coefficient associated with a treatment dummy in outcome regression (1), where the dependent variable is the average growth rate of the corresponding export performance measure indicated in the column header. All the covariates used to estimate the propensity score, including main export product-destination fixed effects, are also included in the estimated regression. Number of exported products is defined at the HS 8-digit level. The sample of firms used in these estimations consists of 2,027 (non-singleton) firms with positive export sales in both 2015 and 2017. There are 122 firms that did not receive the EFS subsidy in 2015 but did so in 2016 and/or 2017 (treated firms). Standard errors in parenthesis ***, significant at the 1% level; **, significant at the 5% level; *, significant at the 10% level.

the magnitude of the subsidy on export sales by 25%.¹⁵

A reduction on the interest rate for working capital loans not only has a direct impact on firms' marginal costs, as emphasized by Manova (2013) and Feenstra et al. (2014), but can also induce firms to increase the scale of their operation by weakening the need to selfinsure against negative shocks (Arellano et al., 2019). In terms of its magnitude, the strong impact on exports that we estimate for EFS is in line with the existing empirical evidence evaluating this type of subsidies. Zia (2008) finds that when cotton yarn was included in EFS's negative list in 2001, affected producers saw their exports decline by 31% vis-à-vis

¹⁵Table C.2 presents the estimated coefficients of the pre-treatment growth rates of export outcomes in the outcome regressions reported in column (1)—i.e. when the dependent variable is the average growth rate of export value. Similarly to what we have reported in the estimation of the selection to treatment model, these estimates indicate that pre-treatment export performance does not have significant explanatory power in explaining the effect of export finance support subsidies on the growth rate of export sales.

	Export	#	#
	value	destinations	products
	(1)	(2)	(3)
OLS	0.219^{***}	0.181^{***}	0.095***
	(0.043)	(0.027)	(0.034)
Inverse probability (IPW)	0.226^{***}	0.190^{***}	0.103^{***}
	(0.064)	(0.035)	(0.039)
Propensity score (PSM)	0.311^{***}	0.182^{***}	0.078
	(0.089)	(0.044)	(0.065)
Mahalanobis matching (NNM)	0.208***	0.162***	0.140***
	(0.077)	(0.047)	(0.052)
Average growth rate			
of treated firms	-0.037	0.041	-0.007

Table 7: Average treatment effect of Long-Term Financing Facility for Machinery & Equipment (LTFF) on the average growth rate of export outcomes

Each entry in the table reports the average treatment effect on the treated firms that made use of LTFF i.e. the estimated coefficient associated with the treatment dummy in outcome regression (1), where the dependent variable is the average growth rate of the corresponding export performance measure indicated in the column header. All the covariates used to estimate the propensity score, including main export productdestination fixed effects, are also included in the estimated regression. Number of exported products is defined at the HS 8-digit level. The sample of firms used in these estimations consists of 745 (non-singleton) firms with positive export sales in both 2015 and 2017. There are 71 firms that did not receive the LTFF subsidy in 2015 but did so in 2016 and/or 2017 (treated firms). Standard errors in parenthesis ***, significant at the 1% level; **, significant at the 5% level; *, significant at the 10% level.

firms exporting non-yarn textiles. Akgündüz et al. (2018) in turn, find that firms that use the export rediscount credit program offered by the Central Bank of Turkey—an interest rate subsidy to working capital loans similar to EFS both in its scope and the magnitude of its outlays—increased their export sales by 65% following a substantial increase in the program's expenditure. The fact that the subsidy provided under EFS affects primarily the intensive margin of exports, echoes the findings of Paravisini et al. (2015) for Peruvian exporters affected by the capital flow reversals that hit the banks they borrowed from during the global financial crisis of 2008. The more muted response along the extensive margin for firms receiving EFS could be indicative of diseconomies of scope or that market access costs are not highly responsive to a lower cost of working capital.

Table 7 presents the estimates for the average impact of LTFF on export performance.

We find that using LTFF has a large and positive effect on firms' export performance, both along the intensive and extensive margins too. More precisely, using LTFF results in an increase of 20-31 percentage points in the growth rate of exports for treated firms relative to the comparison group. With respect to the extensive margin, and similarly to EFS, treatment effects are stronger for the number of destinations served than for the number of products that firms export. The relaxation of credit frictions brought about by the use of LTFF—particularly in a country with low levels of financial development and prone to crises like Pakistan—could enable exporters to accelerate their accumulation of physical capital and increase their sales, consistent with the findings of Rho and Rodrigue (2016), Brooks and Dovis (2020), Leibovici (2021) and Alessandria et al. (2021).

Tables B.4 and B.5 in the online appendix report the treatment effect estimates for each subsidy without including main export product-destination fixed effects in the estimating equations. While qualitatively the results are similar to the benchmark estimates we have just discussed, it is noting that the magnitude of the treatment effects are sensitive to the inclusion or not of fixed effects—thus highlighting the importance of controlling for potential demand shocks that can confound the effect of the subsidies.

It is helpful to put the evaluation of the impact of export finance support schemes in context by comparing it to the estimated impact of other export promotion policies in developing countries. In contrast to our findings, the literature finds that these instruments act primarily through the extensive margin and have a more limited impact on export sales. This could be due to the fact that several interventions undertaken by export promotion agencies such as offering logistic help to meet foreign buyers, or providing market research and information on customs clearance procedures, tend to lower product- and market-access fixed costs (Álvarez and Crespi, 2000; Volpe Martineus and Carballo, 2008, 2010a; Cadot et al., 2015), while reducing the user cost of capital has a large and direct effect on firms' marginal costs and sales instead. Another critical dimension in which the export finance schemes offered by SBP differ from other promotion instruments analyzed in the literature

is in terms of their scale. Defever et al. (2020), for instance, find that when subsidy rates are not very high (like in the case of a 1-2% ad-valorem cash subsidy offered in Nepal they study), benefitting firms do not drastically alter the value of their exports. This is certainly not the case for EFS and LTFF; the value of loans refinanced under these schemes is so large that the attractive provisions they offer to exporters are likely to be a strong driving force in explaining the large positive effects we find they produce in the value of treated firms' exports.¹⁶

Since EFS and LTFF are both available to firms exporting only certain products, we investigate the effect of subsidies on export outcomes according to product eligibility status. The results reported in online appendix D, suggest somewhat surprisingly, that both EFS and LTFF also have a positive impact on exports performance of products that are not eligible to be subsidized. While these findings could reflect selection effects (i.e. firms increasing their exports across the board due a to unobserved demand shock) not fully controlled for by our empirical strategy, an alternative explanation could be that the subsidies might help to relax credit constraints for treated exporters, since several of the non-eligible 'commodities' (particularly for EFS) feature exceptions for specific products that might be quite similar such as basmati and husked rice (see Table A.1 in the online appendix). Nevertheless, it is worth stressing that these results need to be interpreted with caution because they are based on a small number of firms that export both eligible and non-eligible products for a given subsidy.

¹⁶While we observe the value of the subsidized loans that firms obtain when using the schemes, using a continuous measure of treatment exposure to estimate a dose-response function along the lines of Hirano and Imbens (2004) lies beyond the scope of this paper because of the stronger identification assumptions and data requirements that this approach demands. In terms of identification, the use of the generalized propensity score requires that assignment to treatment levels (i.e. the value of the loan that exporters receive) to be as good as random given pre-treatment covariates. This assumption is hard to justify without balance sheet data that allow us to control for the size of loans that a firm could secure from a commercial bank. Furthermore, since this method aims at estimating counterfactual outcomes for each level of treatment, it is crucial to have a larger number of treated observations than what we currently have.

6 A Back-of-the-Envelope Cost-Benefit Analysis

The results presented in the previous section indicate that both EFS and LTFF provide a large boost to the exports of firms that use these programs. Nevertheless, given the considerable reduction in interest rates they offer and the scale of loans they finance, it is important to evaluate how effective they are in achieving these results. This is all the more pressing for Pakistan, a country that systematically runs large fiscal deficits which are, to a large extent financed by direct borrowing from SBP, and that are considered an important source of risk to macroeconomic stability (IMF, 2019).

We conduct a back-of-the-envelope cost-benefit analysis, following the approach used by Cadot et al. (2015) and Munch and Schaur (2018) to infer the value of additional exports 'created' by each scheme and compare this against the direct financial cost of providing the subsidies borne by SBP. It is important, however, to keep in mind that this exercise does not constitute a fully-fledged welfare analysis. The latter would take into consideration important aspects like the extent to which lower interest rates ease credit constraints for exporters, general equilibrium effects—which might be quite substantial given the magnitude of the subsidies offered by the schemes; the distribution of costs and benefits across manufacturing firms that can shape selection into exporting, and the marginal cost of public funds required to administer the schemes, to name a few, that are outside the scope of our analysis.

The results of our 'direct' cost-benefit analysis of EFS and LTFF are summarized in columns (1) and (2) of Table 8 respectively. We use the average of our matching estimates to infer the additional exports generated by each scheme, assuming that all firms that participated in a given subsidy program would experience a higher growth rate of export sales (relative to the counterfactual scenario) equal to the average treatment effect we estimate. More precisely, let $\overline{\beta_S}$ denote the simple average of the matching estimates for the impact of scheme S on the growth rate of export sales for treated firms (the estimates reported in rows 2 to 4 of column (1) of Tables 6 and 7)—i.e. 13.2 and 24.8 percentage points for EFS

	Direc	t cost-benefit	Interna	ul rate of return
	EFS	LTFF	EFS	LTFF
	(1)	(2)	(3)	(4)
Estimated growth rate of exports	13.2%	24.8%		
Growth rate of exports s.t. net benefit=0			2.62%	0.10%
Benefits:				
Exports of first-treated firms in 2017	635m	$3,025\mathrm{m}$		
Additional exports generated by scheme	$79\mathrm{m}$	$667 \mathrm{m}$		
Exports of all treated firms in 2017			8,032m	$6,407\mathrm{m}$
Costs:				
Loans outstanding	108m	$143 \mathrm{~m}$	$3,780\mathrm{m}$	$325\mathrm{m}$
SBP opportunity cost	6.5%	6.5%	6.5%	6.5%
Refinancing rate	1%	4.5%	1%	4.5%
Financial cost SBP	$5.9\mathrm{m}$	$2.9\mathrm{m}$	$2.9\mathrm{m}$	$6.5 \mathrm{m}$
Net benefit:	$72.7\mathrm{m}$	664.6m	$0\mathrm{m}$	$0\mathrm{m}$

Table 8: Cost-benefit analysis of export subsidies

that did not participate in EFS (LTFF) in 2015 but did so in at least one year between 2016 and 2017. The estimated average treatment effect for the of analysis. The financial cost for SBP is calculated as Loans outstanding \times (SBP opportunity cost – Refinancing rate). Net benefits are defined as treated used to calculate the additional exports generated by each scheme is the simple averages of the matching estimates reported in column (1) of the difference between the additional exports generated by a given scheme and its financial cost for SBP. Equation (2) describes how the growth rate All monetary values (i.e. exports, loans outstanding and the financial cost for SBP) are denominated in millions of US dollars. Treated firms are those Table 6 and 7 for EFS and LTFF respectively. Outstanding loans refer to the total loans received by treated firms averaged between 2016 and 2017. The opportunity cost for SBP to raise external funds is the average 6-month treasury bill rate issued by the government of Pakistan over the period of export sales that yields net zero benefits for a given scheme is calculated. and LTFF respectively. Letting $Y_{2017,S} = \sum_{i \in S} y_{i,2017}$ denote the total export sales of firms using scheme $S \in \{\text{EFS}, \text{LTFF}\}$ in 2017, then we calculate the value of exports for treated firms had they not participated in scheme S as $Y_{2017,S}^C = [1 - \overline{\beta_S}/2]/[1 + \overline{\beta_S}/2] \times Y_{2017,S}$.¹⁷ Thus, our results indicate that the additional exports among the treated firms included in the empirical analysis, $(Y_{2017,S} - Y_{2017,S}^C)$, would have been 72.7 and 664.6 million US dollars for EFS and LTFF respectively.

We calculate the financial cost for SBP as the difference between the interest rate at which SBP can raise funds, which we take as the yield of the 6-month treasury bill rate issued by the government of Pakistan (6.5% per annum on average between 2015 and 2017), and the corresponding refinancing rate it charges banks (1% for EFS and 4.5% for LTFF) times the amount of loans outstanding. We chose the yield of the 6-month Treasury bill to represent SBP's opportunity cost of 'printing money' to finance the schemes both because the interest rate charged to exporters on their loans and the refinancing rate charged to commercial banks are linked to the this benchmark rate, and because the 6-month T-bill is the most important debt instrument used by the government (SBP, 2017).¹⁸ Since the value of loans outstanding for the treated firms included in our analysis are 108 and 143 million US dollars for EFS and LTFF respectively, the direct (i.e. without including potential overhead costs of administering the schemes) financial cost of these two schemes for SBP are $(0.065 - 0.01) \times 108 = 5.9$ and $(0.065 - 0.045) \times 143 = 2.9$ million US dollars.

Our analysis suggests that both EFS and LTFF are highly effective instruments to boost exports, as they both deliver a substantial positive direct net benefit—65 and 665 million US dollars respectively. The net impact of LTFF is much higher for LTFF, both because its estimated effect on firms' exports is higher and because the difference between the market

 $^{^{17}}$ Recall that we have used the mid-point growth rate of export sales as our dependent variable in the regressions reported in column (1) of Tables 6 and 7.

¹⁸Since the 2015-2017 period is characterized by low and stable interest rates, the financial cost of the schemes is quite similar regardless of whether we use the discount rate, money market rate, or the yield on Pakistan's sovereign bonds to represent SBP's opportunity cost. These rates range between 6 to 9% on average between 2015 and 2017.

and the subsidized interest rates it offers is smaller than for EFS.¹⁹ These results are in line with the existing literature that has found that the value of additional exports created by different export promotion policies such as the services provided by the Danish Trade Council or matching grants provided to support the development of business plans in Tunisia greatly exceed the direct financial cost of each policy (Munch and Schaur, 2018; Cadot et al., 2015).

One way to gauge the robustness of our analysis is to calculate an internal rate of return for each scheme. In other words, determining the growth rate of export sales that would deliver zero net benefits, based on the *total* value of exports and refinanced loans, rather than only using the values of these variables for the firms included in our estimation—an important consideration given the large number of perennial subsidy users that we have to exclude from our estimation. Letting $Y_{2017,S}^{ALL}$ denote the total value of exports for all firms that used scheme S in 2017 and FinancialCost^{ALL} = Loans outstanding^{ALL} × (SBP opportunity cost – Refinancing rate_S) the corresponding direct financial cost of scheme S for SBP in 2017, then the growth rate of export sales that yields a zero net-benefit for scheme S, β_S^0 , is implicitly defined by:

$$\left[1 - \frac{1 - \beta_S^0/2}{1 + \beta_S^0/2}\right] Y_{2017,S}^{ALL} - \text{FinancialCost}_S^{ALL} = 0.$$
(2)

The solution to equation (2) is reported in the second row of columns (2) and (4) of Table 8. We find that if the subsidy to working capital were to increase the growth rate of exports of treated firms by 2.62 percentage points—which is significantly different than the treatment effect we estimate—then the subsidy would not deliver any net benefits. We find that due to the its substantially lower direct financial cost, LTFF would deliver net benefits as long as it induces firms to increase their exports.

It is worth noting that SBP did not (at least not until quite recently) adjust refinancing rates frequently, even in response to large swings in market rates—which would significantly

¹⁹Redoing the cost-benefit exercise using the more muted average effect of LTFF on export sales implied by the estimates without fixed effects (i.e. that the use of LTFF would have increased the growth rate of exports of treated firms by 8.5 percentage points) reported in Table B.5 in the online appendix, still suggests that this subsidy delivers a substantial net benefit of 244 million US dollars.

increase the financial cost of the schemes. For instance, in order to rein in domestic demand and stave off a balance-of-payments crisis, SBP more than doubled the discount rate from 6.25 to 13.75% over the course of 2018 without changing the refinancing rate of either EFS or LTFF (ADB, 2020). Under these circumstances, exporters using EFS would have to increase their exports by almost 6 percentage points for the subsidy to deliver net gains.

7 Conclusion

In this paper we evaluate the effect of two interest rate subsidies provided to exporters in Pakistan on their exporting performance. the Export Finance Scheme and the Long-Term Finance Facility for Plant & Machinery provide loans at below-market interest rates for exporters to finance investment in working capital and machinery and equipment respectively. A notable feature of these programs is that they finance a substantial share of Pakistan's total exports, and thus are orders of magnitude larger than most instruments of export promotion that have previously been studied in the literature.

We use a matching combined with difference-in-differences empirical strategy to estimate the impact of receiving these subsidies on export performance while addressing the endogenous selection of firms into the programs. Our results show that both EFS and LTFF induce recipient firms to increase their exports sales, with the overall effect being stronger for the subsidy targeted to investment in machinery and equipment. Both programs appear to operate primarily through the intensive margin; however, along the extensive margin, EFS only has significant effects in terms of the number of destinations that exporters serve, while using LTFF has a significantly positive impact on both the number of exported products and destinations. A back-of-the-envelope cost-benefit analysis suggests that both programs are effective in increasing exports given their direct financial cost.

References

- AKGÜNDÜZ, Y. E., S. H. KAL, AND H. TORUN (2018): "Do subsidized export loans increase exports?" *The World Economy*, 41, 2200–2215.
- ALESSANDRIA, G., H. CHOI, AND K. J. RUHL (2021): "Trade adjustment dynamics and the welfare gains from trade," *Journal of International Economics*, 131, 103458.
- ÁLVAREZ, R. E. AND G. T. CRESPI (2000): "Exporter performance and promotion instruments: Chilean empirical evidence," *Estudios de Economía*, 27, 225–241.
- AMITI, M. AND D. E. WEINSTEIN (2011): "Exports and Financial Shocks," The Quarterly Journal of Economics, 126, 1841–1877.
- ARELLANO, C., Y. BAI, AND P. J. KEHOE (2019): "Financial Frictions and Fluctuations in Volatility," *Journal of Political Economy*, 127, 2049–2103.
- ASIAN DEVELOPMENT BANK AND ISLAMIC DEVELOPMENT BANK (2020): Pakistan: Reviving Growth through Competitiveness, Manila.
- AUBOIN, M. (2007): "Boosting trade finance in developing countries: What link with the WTO?" WTO Staff Working Papers ERSD-2007-04, World Trade Organization (WTO), Economic Research and Statistics Division.
- BECK, T., A. DEMIRGÜÇ-KUNT, AND V. MAKSIMOVIC (2008): "Financing patterns around the world: Are small firms different?" Journal of Financial Economics, 89, 467–487.
- BLUM, B. S., S. CLARO, AND I. J. HORSTMANN (2013): "Occasional and perennial exporters," *Journal of International Economics*, 90, 65–74.
- BLUNDELL, R. AND M. COSTA DIAS (2009): "Alternative Approaches to Evaluation in Empirical Microeconomics," *Journal of Human Resources*, 44, 565–640.
- BRICONGNE, J.-C., L. FONTAGNÉ, G. GAULIER, D. TAGLIONI, AND V. VICARD (2012): "Firms and the global crisis: French exports in the turmoil," *Journal of International Economics*, 87, 134–146.
- BROOKS, W. AND A. DOVIS (2020): "Credit market frictions and trade liberalizations," *Journal of Monetary Economics*, 111, 32–47.
- CADOT, O., A. M. FERNANDES, J. GOURDON, AND A. MATTOO (2015): "Are the benefits of export support durable? Evidence from Tunisia," *Journal of International Economics*, 97, 310–324.
- CAGGESE, A. AND V. CUÑAT (2013): "Financing Constraints, Firm Dynamics, Export Decisions, and Aggregate Productivity," *Review of Economic Dynamics*, 16, 177–193.
- CALIENDO, M. AND S. KOPEINIG (2008): "Some Practical Guidance for the Implementation of Propensity Score Matching," *Journal of Economic Surveys*, 22, 31–72.

- CHÁVEZ, J. F., A. C. NOVELLI, AND M. PEREZ LEÓN (2020): "Export Subsidies in Emerging Markets During the Great Trade Collapse," *Economics Bulletin*, 40, 1879–1892.
- CHOR, D. AND K. MANOVA (2012): "Off the cliff and back? Credit conditions and international trade during the global financial crisis," *Journal of International Economics*, 87, 117–133.
- DEFEVER, F., J.-D. REYES, A. RIAÑO, AND M. E. SÁNCHEZ-MARTÍN (2019): "Special Economic Zones and WTO Compliance: Evidence from the Dominican Republic," *Economica*, 86, 532–568.
- DEFEVER, F., J.-D. REYES, A. RIAÑO, AND G. VARELA (2020): "All these worlds are yours, except India: The effectiveness of cash subsidies to export in Nepal," *European Economic Review*, 128, 103494.
- DEMIR, B., T. K. MICHALSKI, AND E. ORS (2017): "Risk-Based Capital Requirements for Banks and International Trade," *The Review of Financial Studies*, 30, 3970–4002.
- FEENSTRA, R. C., Z. LI, AND M. YU (2014): "Exports and Credit Constraints under Incomplete Information: Theory and Evidence from China," *The Review of Economics* and Statistics, 96, 729–744.
- FELBERMAYR, G. J. AND E. YALCIN (2013): "Export Credit Guarantees and Export Performance: An Empirical Analysis for Germany," *The World Economy*, 36, 967–999.
- FERNANDES, A. M., C. FREUND, AND M. D. PIEROLA (2016): "Exporter behavior, country size and stage of development: Evidence from the Exporter Dynamics Database," *Journal of Development Economics*, 119, 121–137.
- FLEISIG, H. AND C. HILL (1984): "The Benefits and Costs of Official Export Credit Programs," in *The Structure and Evolution of Recent U.S. Trade Policy*, ed. by R. E. Baldwin and A. O. Kruger, Chicago, IL: University of Chicago Press, 321–358.
- GÖRG, H., M. HENRY, AND E. STROBL (2008): "Grant Support and Exporting Activity," The Review of Economics and Statistics, 90, 168–174.
- GREENAWAY, D., A. GUARIGLIA, AND R. KNELLER (2007): "Financial factors and exporting decisions," *Journal of International Economics*, 73, 377–395.
- HIRANO, K. AND G. W. IMBENS (2004): "The Propensity Score with Continuous Treatments," in Applied Bayesian Modeling and Causal Inference from Incomplete-Data Perspectives, ed. by A. Gelman and X.-L. Meng, John Wiley & Sons, Ltd, chap. 7, 73–84.
- IMBENS, G. W. AND D. B. RUBIN (2015): Causal Inference for Statistics, Social, and Biomedical Sciences: An Introduction, Cambridge University Press.
- INTERNATIONAL MONETARY FUND (2019): "Pakistan Country Report," Country Report 19/212.

KHWAJA, A. I. AND A. MIAN (2005): "Do Lenders Favor Politically Connected Firms? Rent Provision in an Emerging Financial Market," *The Quarterly Journal of Economics*, 120, 1371–1411.

(2008): "Tracing the Impact of Bank Liquidity Shocks: Evidence from an Emerging Market," *American Economic Review*, 98, 1413–1442.

- KOHN, D., F. LEIBOVICI, AND M. SZKUP (2016): "Financial Frictions and New Exporter Dynamics," *International Economic Review*, 57, 453–486.
 - —— (2022a): "Financial Frictions and International Trade: A Review," Oxford Research Encyclopedia of Economics and Finance.
- (2022b): "No Credit, No Gain: Trade Liberalization Dynamics, Production Inputs, and Financial Development," *International Economic Review*, 64, 809–836.
- LEIBOVICI, F. (2021): "Financial Development and International Trade," Journal of Political Economy, 129, 3405–3446.
- MANOVA, K. (2013): "Credit Constraints, Heterogeneous Firms, and International Trade," *Review of Economic Studies*, 80, 711–744.
- MCKENZIE, D. (2021): "What do you need to do to make a matching estimator convincing? Rhetorical vs statistical checks," https://blogs.worldbank.org/impactevaluations/ what-do-you-need-do-make-matching-estimator-convincing-rhetorical-vs-statistical? cid=SHR_BlogSiteShare_EN_EXT, Accessed: November 21, 2021.
- MOSER, C., T. NESTMANN, AND M. WEDOW (2008): "Political Risk and Export Promotion: Evidence from Germany," *The World Economy*, 31, 781–803.
- MUNCH, J. AND G. SCHAUR (2018): "The Effect of Export Promotion on Firm-Level Performance," *American Economic Journal: Economic Policy*, 10, 357–387.
- MUÛLS, M. (2015): "Exporters, importers and credit constraints," Journal of International Economics, 95, 333–343.
- PARAVISINI, D., V. RAPPOPORT, P. SCHNABL, AND D. WOLFENZON (2015): "Dissecting the Effect of Credit Supply on Trade: Evidence from Matched Credit-Export Data," *Review of Economic Studies*, 82, 333–359.
- PETERSEN, M. A. AND R. G. RAJAN (1997): "Trade Credit: Theories and Evidence," *The Review of Financial Studies*, 10, 661–691.
- REIS, J. G. AND D. TAGLIONI (2013): "Determinants of export growth at the extensive and intensive margins : evidence from product and firm-level data for Pakistan," Policy Research Working Paper Series 6341, The World Bank.
- RHO, Y. AND J. RODRIGUE (2016): "Firm-Level Investment And Export Dynamics," International Economic Review, 57, 271–304.

- RIAÑO, A. (2011): "Exports, Investment and Firm-level Sales Volatility," Review of World Economics (Weltwirtschaftliches Archiv), 147, 643–663.
- STATE BANK OF PAKISTAN (2017): State Bank of Pakistan Annual Report 2017-16, State Bank of Pakistan.
- VAN BIESEBROECK, J., J. KONINGS, AND C. VOLPE MARTINCUS (2016): "Did export promotion help firms weather the crisis?" *Economic Policy*, 31, 653–702.
- VAN BIESEBROECK, J., E. YU, AND S. CHEN (2015): "The impact of trade promotion services on Canadian exporter performance," *Canadian Journal of Economics*, 48, 1481– 1512.
- VOLPE MARTINCUS, C. (2010): Odyssey in International Markets: An Assessment of the Effectiveness of Export Promotion in Latin America and the Caribbean, Washington, DC: Inter-American Development Bank.
- VOLPE MARTINCUS, C. AND J. CARBALLO (2008): "Is export promotion effective in developing countries? Firm-level evidence on the intensive and the extensive margins of exports," *Journal of International Economics*, 76, 89–106.
 - (2010a): "Beyond the average effects: The distributional impacts of export promotion programs in developing countries," *Journal of Development Economics*, 92, 201–214.
- (2010b): "Export Promotion: Bundled Services Work Better," *The World Economy*, 33, 1718–1756.
- WOOLDRIDGE, J. M. (2007): "Inverse probability weighted estimation for general missing data problems," *Journal of Econometrics*, 141, 1281–1301.
- WORLD BANK (2021): Pakistan Development Update: Reviving Exports, Washington, DC: World Bank.
- WORLD ECONOMIC FORUM (2016): "The Global Enabling Trade Report," World Economic Forum and the Global Alliance for Trade Facilitation.
- WORLD TRADE ORGANIZATION (2019): *Trade Policy Review of Bangladesh*, Geneva: World Trade Organization.
- ZIA, B. (2008): "Export incentives, financial constraints, and the (mis)allocation of credit: Micro-level evidence from subsidized export loans," *Journal of Financial Economics*, 87, 498–527.

Evaluating the Impact of Export Finance Support on Firm-level Export Performance: Evidence from Pakistan

Fabrice Defever, Alejandro Riaño and Gonzalo Varela

Online Appendix — Not for Publication

A Products included in EFS negative list and for which LTFF is available

The list of sectors eligible to obtain finance under LTFF is presented in Table A.2. The list of sectors can also be found here: https://www.sbp.org.pk/mfd/2007/Encl_C7.pdf.

The negative list of products that are excluded from EFS was last updated on May 7, 2011. There are 21 commodities included in the list (with main harmonized codes in parenthesis) presented in Table A.1. The negative list can also be found here: https://www.sbp.org.pk/incentives/efs/efs-negative.pdf.

Commodity	Main HC
Raw cotton	52.01, 52.02, 52.03, 1404.20
All types of yarn	$52.05,\ 52.06,\ 5207.1000,\ 5207.9000$
Mutton and beef other than frozen	02.01, 02.02, 02.03, 02.04, 02.06, 02.08, 16.01, 16.02 excl. 0208.20
Petroleum products	$27.08,\ 27.10,\ 27.11,\ 27.12,\ 27.13,\ 27.14,\ 27.15$
	excl. 2710.0011, 2712.1000, 2712.2000
Crude vegetable materials n.e.s.	$06.01,\ 06.02,\ 12.11,\ 12.13,\ 12.14,\ 13.01,\ 13.02,\ 14.01,\ 14.02,\ 14.03,\ 14.04$
	excl. rose buds, sassafrass leaves, guar gum Extract, asafoetida hing
	Lithospermum Vestitum, henna powder
Wool & animal hair	05.02, 05.03, 51.01, 51.02, 51.03, 51.04, 51.05, excl. wool tops
Crude animal material	$04.07,\ 04.08,\ 05.04,\ 05.05,\ 05.06,\ 05.07,\ 05.08,\ 05.09,\ 05.10,\ 05.11$
	excl. animal casings, fatends, bones
All grains incl. grain flour	$10.01,\ 10.02,\ 10.03,\ 10.04,\ 10.05,\ 10.06,\ 10.07,\ 10.08,\ 11.01,\ 11.02,\ 11.03,\ 11.04,\ 11.06$
	excl. packeted irri, basmati, parboiled, white/brown rice, bulk brown rice
Stone, sand and gravel	$25.05,\ 25.06,\ 25.07,\ 25.14,\ 25.15,\ 25.16,\ 25.17,\ 25.20,\ 25.21$
	excl. onyx, marble, granite tiles/slabs
Waste & scrap of all kinds	$23.03,\ 23.04,\ 23.05,\ 23.06,\ 23.07,\ 23.08,\ 26.18,\ 26.19,\ 26.20,\ 26.21,\ 39.15$
	41.10, 47.07, 55.05, 57.03, 71.12, 72.14, 74.04, 76.02, 78.02, 79.02, 80.02
	excl. rice gluten, polyethylene terephthalate (PET) resin
Fertilizer crude	25.10, 3101.00, 3102.50, 3104.10
Jewellery	exported under the entrustment scheme (excl. gold jewellery)
Live animals	$01.01, \ 01.02, \ 01.03, \ 01.04, \ 01.05, \ 01.06$
	excl. hatching eggs, day old chicks
Hides & skins	41.01, 41.02, 41.03
Leather wet blue	41.04, 41.05, 41.06, 41.07, 4104.21 to 4104.29, 4105.11 to 4105.19, 4106.11 to 4106.19, 4107.21
Crude minerals	$25.01,\ 25.02,\ 25.03,\ 25.04,\ 25.07,\ 25.08,\ 25.09,\ 25.11,\ 25.12,\ 25.18,\ 25.19,\ 25.22$
	25.25, 25.26, 25.27, 25.28, 25.29, 25.30, 27.01, 27.02, 27.03. excl. refined, treated salt
Antiques	90.70
All metal ores	$26.01, \ 26.02, \ 26.03, \ 26.04, \ 26.05, \ 26.06, \ 26.07, \ 26.08, \ 26.09, \ 26.10, \ 26.11, \ 26.12, \ 26.13$
	26.14, 26.15, 26.16, 26.17, 74.01, 75.01 excl. magnesite, blister copper
Fur skins	43.01, 43.02, 43.03
Wood in rough or squared	$44.01,\ 44.02,\ 44.03,\ 44.04,\ 44.05,\ 44.06,\ 44.07,\ 44.08,\ 44.09$
Bleached/unbleached cloth	

Table A.2: Sectors for which LTFF is available

Sectors for which LTFF is available Textiles and garments Rice processing Leather and leather products Sports goods Carpets and wools Surgical instruments Fisheries Poultry and meat Processing of fruits and vegetables IT software and services Marble and granite cutting Gems and jewellery cutting Engineering goods Electrical generators Ethanol Pharmaceutical products Regeneration of textile waste **Glass** production Dairy and soda ash production

B Results without Main Export Product-Destination Fixed Effects

In this section we present the estimates from the selection into treatment models (Table B.1), matching diagnostics (Tables B.2 and B.3) and estimates of average treatment effects for EFS (Table B.4) and LTFF (Table B.5) when we do *not* include main export product-destination fixed effects in our estimating equations.

The estimates for the selection model remain very similar to those reported in Table 3, even though the sample of observations used to estimate the probit models is substantially larger due to the fact that we do not have to exclude singletons. Matching procedures also help to substantially reduce the extent of pre-treatment covariate imbalance.

The treatment effect estimates for the impact of EFS are qualitatively and quantitatively quite similar to the ones reported in the main body of the table. The most important distinction between the results without fixed effects and the benchmark estimates, is that the effect of LTFF is notably smaller in the former. The comparison of estimates with and without fixed effects suggests that the covariates included in our estimating equation were not sufficient to define a comparison group of exporters that had a similar composition of their exports across products and destinations as those of treated firms. Ultimately, the inclusion of the fixed effects allows us to carry out a cleaner comparison to estimate the effect of the subsidies.

	EFS	LTFF
	(1)	(2)
Share of exports in EFS	-0.076	
Negative list	(0.144)	
Share of exports in LTFF		0.767^{***}
Eligible list		(0.183)
Log export value	0.204^{***}	0.547***
	(0.024)	(0.056)
Log # buyers	-0.011	0.004
	(0.052)	(0.071)
Log # destinations	0.090	0.154^{*}
	(0.062)	(0.090)
Log # products	0.007	-0.163**
	(0.043)	(0.064)
Δ Log export value	-0.001	0.101
	(0.050)	(0.101)
$\Delta \text{ Log } \# \text{ buyers}$	-0.009	-0.134
	(0.083)	(0.156)
Δ Log # destinations	-0.048	0.037
	(0.103)	(0.189)
$\Delta \text{ Log } \# \text{ products}$	-0.017	0.426^{***}
	(0.072)	(0.133)
Importer status	-0.149*	0.413^{*}
	((0.084))	(0.223)
HS4 \times destination fixed effects	N	ю
Observations	8,135	$8,\!135$
Excluding singletons	8,135	$8,\!135$
Pseudo R-squared	0.117	0.465
Joint significance test (p-value)	0.00	0.00

Table B.1: First-stage probit for the probability of participating in an export finance scheme

The table reports the coefficients of probit models estimated among the set of firms observed in 2015. The dependent variable in column (1) [(2)] takes the value 1 if a firm that did not participate in EFS [LTFF] in 2015 uses EFS [LTFF] in 2016 and/or 2017 and 0 otherwise. All covariates are measured in 2015 (Δ denotes the difference between the value of a given variable in 2015 relative to the value in 2014). Standard errors are in parenthesis. *, **, *** indicate significance at the 10%, 5% and 1% levels, respectively.

quality
matching
of
Indicators
5
Ц
Table

	\mathbf{S} tan	dardize	a unter	ences		Variar	ICE LAUIC	<u> </u>
	Raw	IPW	PSM	NMM	Raw	IPW	PSM	NMM
Panel A: EFS								
Share of exports in EFS in Negative list	-0.47	0.01	-0.10	0.05	0.56	1.06	0.88	1.13
Log export value	1.14	-0.08	-0.04	0.17	0.59	0.57	0.91	1.04
Log # buyers	0.85	-0.06	0.08	0.12	0.84	0.60	0.76	1.07
Log # destinations	0.70	-0.04	0.18	0.07	1.16	0.83	0.92	1.11
$Log \ \# \ products$	0.47	-0.02	-0.01	0.04	1.11	0.91	1.06	1.03
Δ Log export value	0.05	-0.00	-0.14	0.05	0.44	0.70	0.65	1.25
$\Delta \operatorname{Log} \# \operatorname{buyers}$	0.03	0.00	0.01	0.01	0.71	0.99	0.79	1.26
$\Delta \operatorname{Log} \# \operatorname{destinations}$	0.04	0.01	-0.03	0.00	0.97	1.32	1.11	1.33
$\Delta \operatorname{Log} \# \operatorname{products}$	0.01	-0.01	-0.02	-0.04	0.59	0.82	0.86	1.12
Importer status	0.34	-0.03	0.07	0.00	1.23	1.00	1.00	1.00
Panel B: LTFF								
Share of exports of LTFF in eligible list	0.63	0.05	0.03	0.00	0.62	0.91	0.97	1.00
Log export value	2.58	-0.02	0.02	0.37	0.43	0.94	1.05	1.05
Log # buyers	1.91	0.09	0.07	0.17	1.04	1.04	1.40	1.08
Log # destinations	1.73	0.01	0.05	0.10	1.09	0.97	1.48	1.09
$Log \ \# \ products$	1.10	0.13	0.25	0.11	1.05	1.10	1.14	1.27
Δ Log export value	0.05	-0.12	0.21	0.05	0.45	0.46	4.59	1.02
$\Delta \operatorname{Log} \# \operatorname{buyers}$	0.04	-0.02	0.26	0.01	0.46	1.11	1.81	1.55
$\Delta \operatorname{Log} \# \operatorname{destinations}$	0.13	-0.06	0.12	0.03	0.44	1.04	1.12	1.73
$\Delta \operatorname{Log} \# \operatorname{products}$	0.20	-0.04	0.15	0.17	0.28	0.58	0.73	1.24
Importer status	1.60	0.00	-0.12	0.00	0.20	0.98	1.95	1.00

of covariate X_k in the treatment and comparison groups respectively and $s_{k,1}^2$ and $s_{k,0}^2$ are the sample variances of covariate X_k in the treatment and comparison groups respectively. The variance ratio is defined as $VR = s_{k,1}^2/s_{k,0}^2$. IPW stands for inverse probability weighting, PSM stands for propensity score matching weighting and NNM for Mahalanobis matching weighting. The standardized difference for each covariate X_k is given by $SD_k = (\overline{X}_{k,1} - \overline{X}_{k,0})/\sqrt{(s_{k,1}^2 + s_{k,0}^2)/2}$, where $\overline{X}_{k,1}$ and $\overline{X}_{k,0}$ denote the sample mean

EFS		IPW	PSM	NNM
pseudo R-squared	0.12	0.00	0.02	0.01
Joint significance test (p-value)	0.00	1.00	0.68	0.92
LTFF		IPW	PSM	NNM
pseudo R-squared	0.47	0.01	0.04	0.06
Joint significance test (p-value)	0.00	0.95	0.74	0.28

Table B.3: Joint significance and pseudo R-squared of treatment status model

The table reports the pseudo R-squared and the p-value associated with the chi-squared joint significance test from running the probit model of the probability of participating in each export finance scheme, and the same statistics when the model is estimated using only the treated and relevant comparison firms. IPW stands for inverse probability weighting, PSM stands for propensity score matching weighting and NNM for Mahalanobis matching weighting.

Table B.4: Average treatment effect of Export Finance Scheme (EFS) on the average growth rate of export outcomes—excluding main export product-destination fixed effects

Export	#	#
value	destinations	products
(1)	(2)	(3)
0.198***	0.077***	0.040*
(0.034)	(0.020)	(0.022)
0.167^{***}	0.061^{***}	0.026
(0.035)	(0.020)	(0.022)
0.201^{***}	0.053^{**}	0.051^{*}
(0.042)	(0.027)	(0.029)
0.171^{***}	0.055^{**}	0.050^{*}
(0.048)	(0.028)	(0.029)
0.032	0.031	-0.020
	$\begin{array}{c} \text{Export} \\ \text{value} \\ (1) \\ 0.198^{***} \\ (0.034) \\ 0.167^{***} \\ (0.035) \\ 0.201^{***} \\ (0.042) \\ 0.171^{***} \\ (0.048) \\ \end{array}$	Export $\#$ valuedestinations(1)(2) 0.198^{***} 0.077^{***} (0.034) (0.020) 0.167^{***} 0.061^{***} (0.035) (0.020) 0.201^{***} 0.053^{**} (0.042) (0.027) 0.171^{***} 0.055^{**} (0.048) (0.028)

Each entry in the table reports the average treatment effect on the treated firms that made use of EFS—i.e. the estimated coefficient associated with a treatment dummy in outcome regression (1), where the dependent variable is the average growth rate of the corresponding export performance measure indicated in the column header. All the covariates used to estimate the propensity score are also included in the estimated regression. The regressions presented in this table do *not* include main export product-destination fixed effects in the estimation of the treatment and outcome equations. Number of exported products is defined at the HS 8-digit level. The sample of firms used in these estimations consists of 8,135 firms with positive export sales in both 2015 and 2017. There are 143 firms that did not receive the EFS subsidy in 2015 but did so in 2016 and/or 2017 (treated firms). Standard errors in parenthesis ***, significant at the 1% level; **, significant at the 10% level.

	Export	#	#
	value	destinations	products
	(1)	(2)	(3)
OLS	0.217***	0.117^{***}	0.055*
	(0.027)	(0.019)	(0.028)
Inverse probability (IPW)	0.079^{**}	0.053^{**}	-0.011
	(0.031)	(0.022)	(0.031)
Propensity score (PSM)	0.080^{*}	0.056^{*}	-0.010
	(0.047)	(0.031)	(0.035)
Mahalanobis matching (NNM)	0.095^{**}	0.060^{**}	-0.011
	(0.047)	(0.029)	(0.044)
Average growth rate			
of treated firms	-0.053	0.026	-0.026

Table B.5: Average treatment effect of Long-Term Financing Facility for Machinery & Equipment (LTFF) on the average growth rate of export outcomes—excluding main export product-destination fixed effects

Each entry in the table reports the average treatment effect on the treated firms that made use of LTFF—i.e. the estimated coefficient associated with the treatment dummy in outcome regression 1, where the dependent variable is the average growth rate of the corresponding export performance measure indicated in the column header. All the covariates used to estimate the propensity score are also included in the estimated regression. The regressions presented in this table do *not* include main export product-destination fixed effects in the estimation of the treatment and outcome equations. Number of exported products is defined at the HS 8-digit level. The sample of firms used in these estimations consists of 8,839 firms with positive export sales in both 2015 and 2017. There are 82 firms that did not receive the LTFF subsidy in 2015 but did so in 2016 and/or 2017 (treated firms). Standard errors in parenthesis ***, significant at the 1% level; **, significant at the 5% level; *, significant at the 10% level.

C Export Performance Growth Rates Prior to Treatment

In this section we provide evidence suggesting that treated and control firms were experiencing similar export growth trajectories prior to the former group of firms receiving a given export finance subsidy.

Table C.1 reports the effect of growth rate in different dimensions of export performance in explaining the probability of a firm receiving export finance support. Column (1) reproduces the estimates for these covariates from the benchmark selection models reported in Table 3, while in columns (2)-(4) we re-estimate the selection model using only the treated and relevant comparison firms (note that all other covariates are included in the estimation but are not reported in the table). The results from this exercise show that both before and after matching, pre-treatment growth rates in export outcomes are not strong drivers of the probability of using export finance subsidies.

EFS	Unweighted	IPW	PSM	NNM
	(1)	(2)	(3)	(4)
Δ Log export value	0.030	0.023	0.565^{**}	-0.005
	(0.084)	(0.112)	(0.242)	(0.244)
Δ Log # buyers	-0.100	0.021	-0.535*	0.307
	(0.127)	(0.166)	(0.305)	(0.335)
Δ Log # destinations	0.040	-0.009	0.069	-0.192
	(0.138)	(0.174)	(0.313)	(0.378)
$\Delta \log \# \text{ products}$	-0.059	-0.090	0.026	0.219
	(0.111)	(0.149)	(0.291)	(0.313)
\mathbf{LTFF}				
Δ Log export value	0.334^{*}	0.013	-0.714*	0.670
	(0.187)	(0.329)	(0.417)	(0.590)
$\Delta \text{ Log } \# \text{ buyers}$	-0.429	0.222	1.339^{**}	-0.767
	(0.272)	(0.483)	(0.677)	(0.827)
Δ Log # destinations	-0.149	0.012	-0.176	0.215
	(0.321)	(0.438)	(0.700)	(0.700)
$\Delta \log \# \text{ products}$	0.690^{***}	0.160	1.469^{**}	0.926
	(0.234)	(0.398)	(0.658)	(0.619)

Table C.1: Effect of pre-treatment growth rates of export performance in the first-stage probit model predicting the probability of participating in a given export finance scheme

The table reports the effect of pre-treatment growth rates across different dimensions of export performance on the probability of participating in a given export finance scheme. Column (1) reproduces the (unweighted) estimates from the treatment selection model reported in Table 3 in the main text. Columns (2)-(4) report the estimates when we re-estimate the selection model using only the treated and relevant comparison firms. Note that all other covariates are included in the estimation but are not reported. IPW stands for inverse probability weighting, PSM stands for propensity score matching weighting and NNM for Mahalanobis matching weighting. Standard errors are in parenthesis. *, **, *** indicate significance at the 10%, 5% and 1% levels, respectively.

Table C.2 reports the estimated coefficients associated to the pre-treatment growth rate of export performance when estimating the outcome model defined in (1) when the dependent variable is the average growth rate of firm-level export sales over the period 2015-2017. Similarly to the results for the selection model, we find that almost all the estimates of the effect of pre-treatment growth rates of export performance have an insignificant effect on the growth rate of export sales, regardless of the specific matching procedure we employ.

EFS	OLS	IPW	PSM	NNM
	(1)	(2)	(3)	(4)
Δ Log export value	-0.041**	0.003	-0.117	0.015
	(0.017)	(0.040)	(0.094)	(0.085)
Δ Log # buyers	-0.005	-0.067	0.041	-0.027
	(0.023)	(0.046)	(0.098)	(0.103)
Δ Log # destinations	-0.020	-0.034	-0.066	-0.269***
	(0.026)	(0.047)	(0.096)	(0.091)
$\Delta \log \# \text{ products}$	0.032	0.006	-0.033	0.101
	(0.022)	(0.043)	(0.105)	(0.087)
LTFF				
Δ Log export value	-0.030	0.094	0.195^{*}	0.125
	(0.031)	(0.075)	(0.098)	(0.144)
Δ Log # buyers	0.052	0.068	-0.078	0.019
	(0.039)	(0.099)	(0.159)	(0.200)
Δ Log # destinations	-0.076	-0.078	-0.044	-0.158
	(0.047)	(0.109)	(0.146)	(0.172)
$\Delta \log \# \text{ products}$	-0.016	0.047	0.090	0.008
	(0.035)	(0.089)	(0.128)	(0.152)

Table C.2: Effect of pre-treatment growth rates of export performance in outcome model

The table reports the estimated coefficients associated with pre-treatment growth rates in export performance in the estimation of outcome regression (1 when the dependent variable is the average growth rate of firmlevel export sales over the period 2015-2017. Note that all other covariates are included in the estimation but are not reported. IPW stands for inverse probability weighting, PSM stands for propensity score matching weighting and NNM for Mahalanobis matching weighting. Standard errors are in parenthesis. *, **, *** indicate significance at the 10%, 5% and 1% levels, respectively.

D Effect of Subsidies on Export Performance According to Product-eligibility Status

EFS and LTFF are both available to firms exporting only certain products. As noted in Section 2, the negative list of products for EFS is relatively narrow (see Table A.1 in Appendix A) compared to LTFF, which is only available to firms exporting products in 20 broadly-defined industries. In this section we explore whether using export finance schemes affects export performance in targeted and non-targeted products.

To do so, we re-estimate both the treatment and outcome equations defined in Section 4 separately for eligible and non-eligible products, keeping in mind that the pre-treatment covariates used to estimate the propensity score are also defined separately for each subset of products and that, just as in the benchmark estimations, we also include main export

product-destination fixed effects in both the treatment and outcome estimating equations.

The results of this exercise are reported in Tables D.1 and D.2. Columns (1)-(3) in each table present the impact of receiving EFS and LTFF respectively on export performance of eligible products. Since a majority of firms receiving subsidies only export eligible products, and among those exporting non-eligible products, these tend to account for a small share of total exports, these results are quite similar to the benchmark estimates presented in the main body of the paper. Columns (4)-(6) of the same tables, present the effect of subsidies on exports of non-eligible products. These estimates indicate that across most performance measures, export finance schemes also appear to have a positive impact on export performance on non-targeted products, although these estimates are less precisely estimated due the fact that a much smaller sample of firms export both eligible and non-eligible products are often only marginally significant.

These results suggest that receiving export finance support also improves export performance of non-targeted products. On the one hand these results could reflect selection effects, i.e. that firms that increase their exports across the board are more likely to apply and obtain export finance support, that have not been fully controlled for with our empirical strategy. Alternatively, they could also reflect the blurred boundaries between certain eligible and noneligible products, which in turn could result in firms receiving subsidies experiencing positive spillovers in their exports of non-eligible products as financial constraints are relaxed. We find, in fact, that the majority of firms that export both eligible and non-eligible products, export some type of rice (e.g. basmati rice) which is eligible for EFS and other grains like rice in the husk, sorghum or maize, which are not. Nevertheless, it is worth noting again, that we only observe a small number of firms exporting both types of products at the same time, and thus, these results need to be interpreted with caution.

Table D.1: Average Treatment Effect of Export Finance Scheme (EFS) on the Average Growth Rate of Export Outcomes

	1	Proto broaco			A TOWARD A TOWARD	
	Export	#	#	Export	#	#
	value	destinations	products	value	destinations	products
	(1)	(2)	(3)	(4)	(5)	(9)
SIO	0.165^{***}	0.074^{***}	0.040	0.160^{*}	0.059	-0.008
	(0.041)	(0.026)	(0.026)	(0.084)	(0.056)	(0.050)
Inverse probability	0.124^{***}	0.058^{**}	0.029	0.158^{*}	0.057	0.000
(IPW)	(0.045)	(0.025)	(0.030)	(060.0)	(0.064)	(0.051)
Propensity score	0.138^{***}	0.087^{***}	0.052	0.241^{***}	0.118^{***}	-0.037
(PSM)	(0.030)	(0.026)	(0.041)	(0.082)	(0.031)	(0.054)
Mahalanobis matching	0.146^{***}	0.083^{**}	0.034	0.174^{*}	0.063	-0.067
(NNM)	(0.055)	(0.035)	(0.039)	(0.093)	(0.078)	(0.070)
Number of firms	1,871	1,871	1,871	185	185	185

included in the estimation have positive export sales in the respective product category (eligible and negative-list) in 2015 and 2017. The sample of variable is the average growth rate of the corresponding export performance measure indicated in the column header. All the covariates used to estimate the propensity score are also included in the estimated regression. Number of exported products is defined at the HS 8-digit level. All firms Each entry in the table reports the average treatment effect on the treated firms that participated in EFS—i.e. the estimated coefficient associated with a given treatment dummy (either all-treated in columns (1)-(3) or first-instance in columns (4)-(6)) in outcome regression (1), where the dependent firms included in the regressions for eligible products reported in columns (1)-(3) include 107 treated firms, while there are 26 treated firms included in the regressions for non-eligible products in columns (4)-(6). Standard errors in parenthesis ***, significant at the 1% level; **, significant at the 5% level; *, significant at the 10% level.

Table D.2: Average Treatment Effect of Long-Term Financing Facility for Machinery & Equipment (LTFF) on the Average Growth Rate of Export Outcomes

		Eligible list		-	Non-eligible lis	9
	Export	#	#	Export	#	#
	value	destinations	products	value	destinations	products
	(1)	(2)	(3)	(4)	(5)	(9)
SIO	0.253^{***}	0.181^{***}	0.091^{**}	0.172^{*}	0.159^{***}	0.034
	(0.046)	(0.028)	(0.035)	(0.097)	(0.054)	(0.071)
Inverse probability	0.232^{***}	0.185^{***}	0.095^{**}	0.219^{*}	0.161^{***}	0.045
(IPW)	(0.067)	(0.037)	(0.046)	(0.122)	(0.059)	(0.072)
Propensity score	0.297^{***}	0.174^{***}	0.108	0.328^{**}	0.163^{***}	0.130^{*}
(PSM)	(0.078)	(0.040)	(0.081)	(0.157)	(0.056)	(0.077)
Mahalanobis matching	0.222^{***}	0.183^{***}	0.136^{**}	0.107	0.142^{*}	-0.002
(NNM)	(0.083)	(0.049)	(0.057)	(0.135)	(0.072)	(0.081)
Number of firms	664	664	664	174	174	174

with a given treatment dummy (either all-treated in columns (1)-(3) or first-instance in columns (4)-(6)) in outcome regression (1), where the dependent variable is the average growth rate of the corresponding export performance measure indicated in the column header. All the covariates Each entry in the table reports the average treatment effect on the treated firms that participated in LTFF—i.e. the estimated coefficient associated All firms included in the estimation have positive export sales in the respective product category (eligible and non-eligible for LTFF) in 2015 and 2017. The sample of firms included in the regressions for eligible products reported in columns (1)-(3) include 65 treated firms, while there are 22 used to estimate the propensity score are also included in the estimated regression. Number of exported products is defined at the HS 8-digit level. treated firms included in the regressions for non-eligible products in columns (4)-(6). Standard errors in parenthesis ***, significant at the 1% level; **, significant at the 5% level; *, significant at the 10% level.

CENTRE FOR ECONOMIC PERFORMANCE Recent Discussion Papers

2026	Andrés Barrios-Fernández Christopher Neilson Seth Zimmerman	Elite universities and the intergenerational transmission of human and social capital
2025	Agnes Norris Keiller	Brexit and investment
2024	Lídia Farré Libertad González Claudia Hupkau Jenifer Ruiz-Valenzuela	Paternity leave and child development
2023	Gabriel M. Ahlfeldt Nikodem Szumilo Jagdish Tripathy	Housing-consumption channel of mortgage demand
2022	Agnes Norris Keiller Tim Obermeier Andreas Teichgraeber John Van Reenen	When trade drives markup divergence. An application to auto markets
2021	Amanda Dahlstrand Nestor Le Nestour Guy Michaels	Online versus in-person services: Effects on patients and providers
2020	Hanwei Huang Kalina Manova Oscar Perelló Frank Pisch	Firm heterogeneity and imperfect competition in global production networks
2019	Jan David Bakker Nikhil Datta	Avenging the tenants: Regulating the middle man's rents
2018	Agnes Norris Keiller Áureo de Paula John Van Reenen	Production function estimation using subjective expectations data

2017	Philippe Aghion Lint Barrage David Hémous Ernest Liu	Transition to green technology along the chain supply
2016	Tom Schmitz Italo Colantone Gianmarco Ottaviano	Regional and aggregate economic consequences of environmental policy
2015	Agnes Norris Keiller Tim Obermeier Andreas Teichgraeber John Van Reenen	An engine of (pay) growth? Productivity and wages in the UK auto industry
2014	Antonin Bergeaud Pierre Cahuc Clément Malgouyres Sara Signorelli Thomas Zuber	The wage of temporary agency workers
2013	Richard Hornbeck Guy Michaels Ferdinand Rauch	Identifying agglomeration shadows: Long-run evidence from ancient ports
2012	Nikhil Datta	Local monopsony power
2011	Hanna Virtanen Mikko Silliman Tiina Kuuppelomäki Kristiina Huttunen	Education, gender and family formation
2010	Giorgio Barba Navaretti Matteo Bugamelli Emanuele Forlani Gianmarco I.P. Ottaviano	It takes (more than) a moment: Estimating trade flows with superstar exporters
2009	Giordano Mion Joana Silva	Trade, skills and productivity

The Centre for Economic Performance Publications Unit Tel: +44 (0)20 7955 7673 Email <u>info@cep.lse.ac.uk</u> Website: <u>http://cep.lse.ac.uk</u> Twitter: @CEP_LSE